Author's response to reviews

Title: A randomised feasibility study of EPA and Cox-2 inhibitor (Celebrex) versus EPA, Cox-2 inhibitor (Celebrex), Resistance Training followed by ingestion of essential amino acids high in leucine in NSCLC cachectic patients-ACCeRT Study.

Authors:

Elaine S Rogers (elaine.rogers@adhb.govt.nz)
Roderick D MacLeod (rd.macleod@auckland.ac.nz)
Joanna Stewart (j.stewart@auckland.ac.nz)
Stephen P Bird (sbird@csu.edu.au)
Justin WL Keogh (jkeogh@bond.edu.au)

Version: 2 Date: 21 October 2011

Author's response to reviews:

Duplicate of Reference number 38 and number 51.